Author's response to reviews

Title: Deltopectoral flap revisited for reconstruction surgery in patients with advanced thyroid cancer: a case report

Authors:

Taro Mikami (zeong3@mac.com)
Shintaro Kagimoto (kagimoto@yokohama-cu.ac.jp)
Yuichiro Yabuki (yu1rou@yokohama-cu.ac.jp)
Kazunori Yasumura (burngy@hotmail.com)
Toshinori Iwai (iwai104oams@yahoo.co.jp)
Jiro Maegawa (maegawaj@yokohama-cu.ac.jp)
Nobuyasu Suganuma (nsuganuma@kcch.jp)
Shohei Hirakawa (hirakawas@nanbu.saiseikai.or.jp)
Katsuhiko Masudo (masudo@urahp.yokohama-cu.ac.jp)

Version: 1 Date: 30 Jul 2017

Author’s response to reviews:

July 30th, 2017

Marco Ettore Allaix
Editor
BMC Surgery

Dear Mr. Allaix

We wish to re-submit the attached Case Report titled “Deltopectoral flap revisited for reconstruction surgery in patients with advanced thyroid cancer: a case report” (BSUR-D-17-00104).

The manuscript has been rechecked and appropriate changes have been made in accordance with the reviewers’ suggestions. For your convenience, the revisions are highlighted in blue in the manuscript.
We thank you and the reviewers for your thoughtful suggestions and insights, which have enriched the manuscript and produced a better and more balanced account of our research. We hope that the revised manuscript is now suitable for publication in your journal.

Sincerely,

Taro MIKAMI

Department of Plastic and Reconstructive Surgery, Yokohama City University Hospital
236-004 Fukuura 3-9, Kanazawa-ku, Yokohama city, Kanagawa-ken, Japan
TEL: +81-45-787-2709
FAX: +81-45-783-5351
E-mail: zeong3@mac.com

Reviewer reports:

Fausto Fama', Ph.D., M.D. (Reviewer 1): Well written and interesting paper.
A careful English language revision needs.
This article is acceptable for publication.
Response: Thank you for your feedback. The manuscript has been checked for language by Editage (www.editage.jp)

Yong Sang Lee (Reviewer 2): This is an interesting case report for advanced thyroid cancer treatment.
There were no recommendations for correction of manuscript.
Response: Thank you for your feedback. We have made a few minor changes in the manuscript based on comments from the Editor.